Author's response to reviews

Title: Association study between SNP rs150689919 in the DNA demethylation gene, TET1, and Parkinson's disease in the Chinese Han population

Authors:

Xin-xin Liao (xinliao_86@126.com)
Zi-xiong Zhan (zhanzixiong@163.com)
Ying-ying Luo (luoyingying08@yahoo.com.cn)
Kai Li (liskai@126.com)
Jun-ling Wang (wjling8002@126.com)
Ji-feng Guo (guojifeng2003@yahoo.com.cn)
Xin-xiang Yan (xxyan1268@yahoo.com.cn)
Kun Xia (xiakun@sklmg.edu.cn)
Bei-sha Tang (bstang7398@163.com)
Lu Shen (shenlu2505@126.com)

Version: 2 Date: 8 June 2013

Author's response to reviews:

The reference style has been revised.